



Working Paper Number 117 April 2007

Cost-Effective Prevention of Diarrheal Diseases: A Critical Review

By Michael Kremer and Alix Peterson Zwane

Abstract

This paper critically reviews the existing research on the cost-effective prevention and treatment of diarrheal diseases, and identifies research priorities in this area aimed at finding ways to reduce the diarrheal disease burden. In contrast to the empirical knowledge base that exists for traditional child health programs to reduce diarrheal morbidity and mortality, evidence on the relative effectiveness and costeffectiveness of various environmental health interventions is limited and subject to significant methodological concerns. There is a limited understanding of the determinants of long-term water and sanitation technology adoption and behavior change at the individual level. Even less is known about how collective action problems in water and sanitation infrastructure maintenance can be overcome. An agenda for future research includes evaluating alternative transmission interruption mechanisms, improving understanding of the determinants of individual-level technology adoption in the water and sanitation sector, and assessing the quality of infrastructure maintenance under different management schemes.

The Center for Global Development is an independent think tank that works to reduce global poverty and inequality through rigorous research and active engagement with the policy community. Use and dissemination of this Working Paper is encouraged, however reproduced copies may not be used for commercial purposes. Further usage is permitted under the terms of the Creative Commons License. The first draft of this paper was prepared under the auspices of the Center for Global Development's Global Health Policy Research Network, with support from the Bill & Melinda Gates Foundation. The views expressed in this paper are those of the author and should not be attributed to the directors or funders of the Center for Global Development.

www.cgdev.org

Cost-effective prevention of diarrheal diseases: A critical review*

This draft: March 2006

Michael Kremer** Harvard University, Brookings Institution, and NBER

Alix Peterson Zwane***
University of California Berkeley and Giannini Foundation

Abstract This paper critically reviews the existing research on the cost-effective prevention and treatment of diarrheal diseases, and identifies research priorities in this area aimed at finding ways to reduce the diarrheal disease burden. In contrast to the empirical knowledge base that exists for traditional child health programs to reduce diarrheal morbidity and mortality, evidence on the relative effectiveness and cost-effectiveness of various environmental health interventions is limited and subject to significant methodological concerns. There is a limited understanding of the determinants of long-term water and sanitation technology adoption and behavior change at the individual level. Even less is known about how collective action problems in water and sanitation infrastructure maintenance can be overcome. An agenda for future research includes evaluating alternative transmission interruption mechanisms, improving understanding of the determinants of individual-level technology adoption in the water and sanitation sector, and assessing the quality of infrastructure maintenance under different management schemes.

Word count of main body: 7300

^{*} This paper is based on a report submitted to the Center for Global Development Global Health Policy Research Network. Anthony Keats, Jessica Leino, Kavita Shukla, Jesse Tack, and Heidi Williams provided valuable research assistance. We thank Jeff Albert, Tom Clasen, Rachel Glennerster, Ranjiv Khush and Ted Miguel for helpful comments and discussion. All errors are the responsibility of the authors.

^{**}Department of Economics, Littauer 207, Harvard University, Cambridge, MA 02138, (617) 495-9145 (t), (617) 495-7730 (f), mkremer@fas.harvard.edu.

^{***} Department of Agriculture and Resource Economics, UC Berkeley, Berkeley, CA 94720, (510) 642-7628 (t), (510) 643-8911 (f), zwane@are.berkeley.edu. Corresponding author.

Summary

Diarrheal diseases kill two million children in poor countries each year. This paper critically reviews the existing research on the cost-effective prevention and treatment of diarrheal diseases, and identifies research priorities in this area, particularly for social scientists, aimed at finding ways to reduce the disease burden.

A series of randomized trials has established the impact and cost-effectiveness of several child health interventions, including breastfeeding, vaccination, oral rehydration therapy, and micronutrient supplementation, in preventing and treating diarrheal diseases. The evidence generated by these trials has played a critical role in encouraging the implementation of these interventions on a large scale, resulting in millions of lives saved.

In contrast to the empirical knowledge base that exists for traditional child health programs to reduce diarrheal morbidity and mortality, the evidence on the relative effectiveness and cost-effectiveness of various environmental health interventions, including water and sanitation hardware and hygiene or health education, is limited and subject to significant methodological concerns. Moreover, we have a very limited understanding of the determinants of long-term water and sanitation technology adoption and behavior change at the individual level. Even less is known about how collective action problems in water and sanitation infrastructure maintenance can be overcome.

We identify a tentative agenda for future research, including:

- 1) Evaluating alternative diarrhea transmission interruption mechanisms—including water and sanitation services as well as hygiene behavior changes—on a comparable basis:
- 2) Understanding the determinants of individual-level technology adoption in the water and sanitation sector;
- 3) Assessing the quality of infrastructure maintenance under different management schemes, thus expanding efforts to improve collective action outcomes in this sector.

Methodologically, several design issues will be critical. First, randomized evaluations are needed in order to distinguish the impact of environmental health interventions from potential confounding factors. Second, samples should be of sufficient size to overcome geographic clustering of health outcomes. Third, trials should be designed to allow estimation of the impact of individual interventions as well as packages of interventions.

Research along these lines would provide policymakers with evidence of "what works" in environmental health that is of similar quality to the existing evidence on child health interventions – evidence which in turn can be translated into policy reforms and, ultimately, into saving lives.

1 Introduction

Diarrheal diseases kill around two million children in poor countries each year (WHO 2002b, Kosek *et al.* 2003). There is widespread agreement that massive reductions in diarrheal mortality will be needed to achieve the Millennium Development Goal (General Assembly of the United Nations 2000) of reducing the under-five child mortality rate by two-thirds (World Bank 2003).

Notwithstanding their ongoing burden, a great deal is known about how to prevent and treat diarrheal diseases. As discussed below, a series of randomized trials have established that several child health interventions—including extending breastfeeding, immunization, oral rehydration therapy (ORT), and micronutrient supplementation—are both effective and cost-effective in treating and preventing diarrhea. The evidence from these trials has played a critical role in encouraging the implementation of these interventions on a large scale, saving millions of lives.

Evidence on the effectiveness and cost-effectiveness of environmental health interventions that could prevent diarrheal diseases is much weaker than in the case of child health interventions like vaccines and ORT. While the underlying biological relationships between water and diarrheal disease are well understood, systematic, controlled comparisons of different approaches are lacking. Based on the existing evidence, it is unclear whether the diarrheal disease control priority in poor countries should be to expand the provision of standard child health interventions such as vaccines, ORT, and micronutrient supplementation, or to complement these interventions with water, sanitation, and health behavior change interventions.

Priorities are also unclear within the water sector itself. New research suggests that rapid declines in the U.S. child mortality rate in the early twentieth century were largely a result of improvements in water quality (Cutler and Miller 2005). Similar large-scale investments in piped water infrastructure are infeasible in very poor countries today, leading some argue for prioritizing new point-of-use systems such as chemical disinfectants (which have been shown to be effective in reducing diarrhea incidence); others for digging deep boreholes; and still others for building many cheap shallow wells that allow a more rapid increase in the supply of water (thus allowing people to more frequently wash dishes, clothes, and themselves) but are more easily subject to contamination. Interventions such as improved hand-washing practices or point-of-use water treatment require voluntary behavior change, but there is little evidence on whether such behavior changes persist in the longer run. Rural water facilities can be long-lived if properly serviced, but they often fall into disrepair quickly due to poor maintenance, and while many different approaches to the organization of maintenance have been advocated, there is little hard evidence on their relative effectiveness. Each class of interventions differs in its ability to impact morbidity and/or mortality outcomes, and likely has very different risk and cost profiles as a result of differences in delivery mechanisms, target groups, and cost-recovery opportunities.

This paper critically reviews the existing research on the cost-effective prevention and treatment of diarrheal diseases in rural areas of developing countries.^{1 2} As we

¹ In this paper we restrict our attention to rural areas because water and sanitation coverage rates, a key determinant of diarrheal disease prevalence, are much lower outside of urban areas both globally and in the poorest regions (WHO 2000). World-wide, 94 percent of urban residents have access to improved water sources, and 86 percent have access to improved sanitation. In rural areas, only 71 percent of the population is covered by improved water and 38 percent is covered by improved sanitation. In Africa, urban water and sanitation coverage rates are about double the fraction of the rural population that has access to improved service. Because coverage rates are so different across settings, relevant policy issues also differ across contexts.

discuss, the available research on the relative effectiveness and cost-effectiveness of various hardware and software investments is subject to significant methodological concerns. In light of this limited evidence, we identify outstanding research needs and discuss existing efforts to improve our understanding of the best means of using scarce resources to tackle diarrheal diseases in poor countries. We argue there is a continuing need for randomized impact evaluations that allow for assessment of the effectiveness and cost-effectiveness of environmental health interventions. Such efforts would provide needed evidence to policymakers on "what works" for water sector interventions in developing countries—evidence that in turn can be translated into policy reforms and, ultimately, into saving lives.

This paper is organized as follows. The remainder of Section 1 provides a brief background on water-related diseases and their transmission (1.1), briefly discusses the role of child health interventions (1.2), and reviews some of the methodological problems with the existing literature on environmental health interventions (1.3). Section 2 reviews the available evidence on the effectiveness of source water quality improvements and sanitation investments, and discusses the need for further work examining alternative means of maintaining these "hardware" investments. Section 3 reviews the available evidence on point-of-use water treatment systems as well as two hygiene behavior modification interventions (health education and hand-washing promotion efforts) and discusses the need for further work examining the long-term effectiveness and cost-

I

In urban settings, improved service quality and efficiency gains for monopoly service providers are typically more important than in rural areas where the challenge of providing access to dispersed communities remains the key concern (Galiani *et al.* 2002, World Bank 2004a).

² There is related work in preparation by the Disease Control Priorities Project at the US National Institutes of Health (*Disease Control Priorities in Developing Countries*, 2nd edition) that will discuss diarrheal diseases (chapter in preparation by Gerald Keusch and Olivier Fontaine) as well as water supply, sanitation, and hygiene promotion (chapter in preparation by Sandy Cairncross and Vivian Valdmanis). Martines et al. (1993) also review existing evidence on the relative cost-effectiveness of various interventions to combat diarrhea.

effectiveness of these individual-level behavior change interventions. Section 4 concludes with a discussion of an agenda for future research.

1.1 Brief background on water-related diseases

Water destined for human contact that is exposed to the environment is a potential source of diarrheal disease. In particular, surface water in less developed countries is often contaminated with pathogens (including bacteria, viruses, and parasites) due to contact with human and livestock waste. Drinking, handling, cooking, and bathing in such microbiologically unsafe water exposes people, and especially young children, to a wide range of health risks including diarrheal diseases. Moreover, the lack of adequate water of any kind reduces the opportunity to wash dishes, clothes, and people and thus contributes to the spread of disease.

One standard method of classifying the disease burden associated with water summarizes the potential health costs of relying on unsafe water according to pathogen transmission path. In this taxonomy, presented in Table 1, diarrheal diseases—the focus of this paper—are spread through fecal-oral transmission and fall into both the waterborne and water-washed categories (White *et al.* 1972, Feachem 1977, Cairncross 1996, International Institute for Environment and Development 2000).

The health cost of diseases transmitted via the fecal-oral route is tremendous, and falls disproportionately on young children. Diarrheal illnesses accounted for at least eight percent of total lost disability-adjusted life years (DALYs) in developing countries in 1990 (Smith *et al.* 1999) and account for perhaps 20 percent of deaths among children under age five (Kosek *et al.* 2003). Not only can acute diarrhea result in severe dehydration, but persistent diarrhea may predispose children to, and exacerbate,

malnutrition (Briend 1990, Schorling *et al.* 1990, Lancet 1991, Guerrant *et al.* 1992). Prospective (though non-randomized) community-based fieldwork in Sub-Saharan Africa summarized by the Child Health Research Project (1998) concludes that diarrhea leads to impaired weight gain, particularly for infants less than one year of age and those not exclusively breast-feeding. Malnutrition is in turn associated with increased risk of death from childhood illnesses (Pelltier *et al.* 1995).³

While the biological relationship between diarrheal morbidity water-washed or water-borne pathogens is well understood, the relationship between diarrheal morbidity and various interventions that interrupt pathogen transmission is less clear. This is in part because, as discussed above, individual behavior, collective action, and complementary technologies can mediate infection.

1.2 The role of child health interventions

Randomized trials have established that several child health interventions—including breastfeeding, immunization, oral rehydration therapy (ORT), and micronutrient supplementation—are both effective and cost-effective in treating and preventing diarrhea (for a review see Hill *et al.* 2004). Exclusive breastfeeding of infants is widely accepted as a means of preventing diarrhea in infants up to six months of age and continued breastfeeding for older children also has protective effects (Raisler *et al.* 1999, Perera *et al.* 1999, WHO Collaborative Study Team 2000). For at least two diarrheal diseases (rotavirus and cholera), many in the public health community believe

-

³ Perhaps because of the endogenous relationship between diarrhea and malnutrition, some observational studies also suggest that early childhood diarrhea is correlated with reduced fitness and cognitive performance for children age six to nine (Guerrant *et al.* 1999, Berkman *et al.* 2002, Niehaus *et al.* 2002), but have not been able isolate a causal effect of diarrheal morbidity. Impact evaluations of programs providing food supplements to primary school children (Chavez and Martinez 1986, Martorell 1993, Pollitt *et al.* 1993) generally show a positive relationship between improved nutrition and schooling and cognitive outcomes.

vaccines have a valuable role to play in preventing these diseases (Glass *et al.* 2004, WHO 2004). ORT appears to have largely been responsible for significant reductions in diarrheal mortality since 1980 (Miller and Hirschorn 1995, Victora *et al.* 1996 and 2000). Micronutrient supplementation, including both therapeutic and preventative supplementation of zinc as well as therapeutic supplementation of vitamin A, has also been found to have positive impacts on diarrheal disease morbidity and mortality (Grotto 2003, Zinc Investigators' Collaborative Group 1999 and 2000, Black 1998, Ramakrishnan and Martorell 1998, Beaton *et al.* 1993). The evidence generated by rigorous evaluations of such health interventions has played a critical role in encouraging the implementation of these interventions on a large scale, saving millions of lives.

Nonetheless, persistently high diarrheal mortality rates show that there is scope for additional preventative investments. These investments may include expanded access to traditional child health programs, but may also potentially include a further class of environmental health interventions. This is because the marginal cost of expanding the child health interventions to uncovered populations may be high, even though the average cost of the interventions are often relatively low. For example, continuous zinc supplementation can be costly if households must be induced to adopt micronutrient "sprinkles" sachets where fortification is infeasible (Zlotkin 2005, USAID 2004a). There is no vaccine against pathogenic *Escherichia coli*, the most common cause of diarrhea in less-developed countries, and vaccines against rotavirus and cholera will take time to roll out. Expanding use of strictly curative ORT, in itself potentially difficult, will reduce diarrheal mortality, but not disease incidence.

Recommendations about the extent to which further efforts to prevent diarrhea should emphasize environmental health investments including diarrheal disease transmission interruption mechanisms like water and sanitation infrastructure and hygiene behavior change are difficult to make, however. The existing literature cannot be used to confidently estimate the relative cost-effectiveness of environmental health interventions.

Existing estimates of the relative cost-effectiveness of environmental health interventions and child health programs to combat diarrheal diseases (summarized in Table 2) are largely based on studies that are subject to significant methodological concerns that we review in section 1.3. ⁴ In addition, important considerations that make environmental health interventions different from child health interventions are typically not taken into consideration in cost-effectiveness analysis (Briscoe 1984, Okun 1988, Varley *et al.* 1998, Meddings *et al.* 2004). As Briscoe (1984) and Okun (1988) emphasize, gains associated with infrastructure provision can extend far beyond mortality and morbidity impacts; women's time may be freed from water transportation duties and thus income-generating activities may be facilitated. In urban and peri-urban areas, improved water supply may also free households from purchasing drinking water from vendors.

⁴ For interventions that impact older children and adults, these figures understate the effectiveness of the intervention. Similarly, for interventions that affect diseases other than diarrhea (like measles), these numbers again understate effectiveness because we report impacts on diarrhea only.

1.3 Methodological concerns with the existing literature on environmental health interventions

In this section, we review several methodological concerns associated with the existing literature on water, sanitation, and hygiene education/behavior change. The shortcomings of much of the research include a lack of a plausible comparison group (1.3.1), clustered sampling and reduced power of statistical tests (1.3.2), the inability to estimate separate impacts of individual interventions (1.3.3), and inadequate means of accounting for externalities (1.3.4). We will refer back to these methodological issues throughout this paper when reviewing the research on various classes of interventions.

1.3.1 Lack of a plausible comparison group

A central shortcoming of the existing literature is its reliance on retrospective, non-randomized approaches (for example, comparing outcomes in villages with wells to outcomes in other nearby villages without wells). Such comparisons are problematic because villages with and without wells may differ along other dimensions that also affect the incidence of diarrheal disease. For instance, a village may have a well because it is better organized or wealthier. These attributes may themselves make residents healthier, whether or not they have a village well, so disentangling the "well effect" from the "wealth effect" is in practice difficult, if not impossible. Controlling for all the important differences between villages in multivariate regression analysis may be difficult, if not impossible. Retrospective strategies are especially suspect in this context due to the considerable variation in diarrhea from year to year, as well as the possibility of underlying trends that may differ across even small geographic regions.

1.3.2 Clustered sampling and reduced power of statistical tests

When a comparison group is included, studies often compare outcomes between households in a small number of treatment villages or communities and those in a small number of comparison villages or communities. Because households within a community are likely to be similar to each other, this clustered sampling reduces the power of statistical tests to determine the existence and size of the treatment effect. The intracluster correlation of outcome measures is critical to assessing the statistical power of treatment effects estimated in clustered samples, yet most studies fail to report this statistic, making it difficult to draw inferences.

To capture the variance in outcome variables across space and time, many localities must be included in both the treatment and comparison groups. While the exact number of clusters required for a study depends on context-specific estimates of the intra-cluster correlation of outcome variables, Esrey (1996) suggests that at least twenty clusters should be included in both treatment and comparison samples for moderate reductions in diarrhea incidence to be detectable with a reasonable degree of power. In practice, widely cited studies often are based on two or three treatment clusters and a similar number of comparison clusters. For example, two of the most widely cited studies on the impact of source water quality improvements (Aziz *et al.* 1990, Huttly *et al.* 1987) each include only five villages in total.

1.3.3 Inability to estimate separate impacts of individual interventions

The relative effectiveness of several interventions (*e.g.* latrine construction, borehole well construction, and hygiene education) cannot be separately estimated when these services are provided together, as has often been the case. In a similar manner, when

evaluating the impact of well construction it is difficult to estimate separate the simultaneous effects of improved water quality at the source and increased water quantity.

From a policy perspective, it is important to be able to distinguish the relative effectiveness of interventions since providing a full package may be prohibitively expensive or difficult (as suggested by the difficulty that Huttley *et al.* [1987] experienced in attempting to roll out a hygiene education effort simultaneously with well construction).⁵

1.3.4 Inadequate means of accounting for externalities

The estimation of the treatment effect may be complicated by the presence of externalities, but studies often fail to account for this possibility. It may be that households near comparison springs, latrines, or wells experience health benefits as a result of the intervention because improved service has reduced diarrhea prevalence in a geographic region. If this is the case, a simple "intention to treat" estimation strategy will underestimate the impact of the intervention. There may also be health externalities among households near the same treatment facility. If this is the case then a "treatment on the treated" estimation strategy is inappropriate (Angrist *et al.* 1996). For example, proximity to a treatment well may affect health through the improved disease environment rather than solely by increasing the probability that the individual will consume better quality water from a protected well. Similar concerns arise regarding community-level hygiene education. Miguel and Kremer (2004) show that cross-cluster

-

⁵ A related concern is that well and latrine location may be difficult to randomize because of social or political concerns. Research strategies in which data is collected at hospitals can run into the similar problems, in that it may be difficult to select data collection centers randomly. See Daniels *et al.* (1990) and Meddings *et al.* (2004) for examples of this approach in case-control settings.

externalities like those that may arise in such cases can be accounted for in estimation by controlling for whether and to what extent individuals assigned to comparison communities live near treatment wells (*i.e.* near areas where the disease environment may improve).

Further research that measures the externalities associated with water and sanitation service provision is needed. Valuing externalities will allow for more accurate comparison of the relative cost-effectiveness of infrastructure investments as compared to interventions that provide strictly private benefits or have more limited impacts on the disease environment. Externalities are particularly important in this context because the average cost of curative approaches to diarrheal disease treatment may be very low and it is often difficult to justify large-scale investments in preventative measures on the basis of existing, admittedly incomplete, cost-effectiveness estimates (Hutton 2001).

2 Source water quality improvements and sanitation provision

In this section, we review the existing epidemiological evidence on the effectiveness of source water quality improvements (2.1) and sanitation provision (2.2) in the prevention of diarrheal diseases. In Section 2.3, we discuss the need for work examining infrastructure maintenance schemes for these types of "hardware" interventions

These transmission interruption mechanisms reduce the probability that surface water will become contaminated with fecal matter. Source water quality improvements (e.g. well construction or spring protection) allow households access to water sources that are less likely to have been contaminated by fecal matter as a result of environmental

exposure.⁶ Some source water quality improvements may also increase the supply of water, thus allowing people to more frequently wash dishes, clothes, and themselves. Latrine provision and use removes fecal matter from the environment, reducing the chances of surface water contamination due to run-off.

2.1 Source water quality improvements

Lack of access to improved water and sanitation services is strongly correlated with poverty. People living in poor rural regions are the most likely to rely on poor-quality water—defined as microbiologically unsafe water contaminated with disease-causing organisms (WHO 2002a). Among individuals living in rural areas of developing countries, about 30 percent (or approximately 926 million people) lack a safe and accessible water supply as defined by the World Health Organization (WHO 2000). The un-served population is most likely to live in Sub-Saharan Africa and South Asia, where 256 million people and 595 million people, respectively, lack access to safe drinking water.^{7 8}

A large body of epidemiological literature investigates the impact of improved water supply and sanitation service provision (often as part of a package of interventions that includes hygiene education) on health outcomes (reviews include Blum and Feachem

⁶ Spring protection clears and seals off the eye, or source, of a spring so that water flows through a pipe rather than seeping from the ground. This is a widely-used technology in Sub-Saharan Africa (Mwami 1995, Lenehan and Martin 1997, UNEP 1998), though it is unsuitable for semi-arid regions (UNEP 1998). While service remains dependent on groundwater levels and groundwater quality, spring protection is intended to protect surface water from contamination.

⁷ These numbers likely understate the proportion of the population actually consuming water of poor quality (WHO 2002a) since the WHO definition of an accessible water supply includes water sources some distance from the home. Reasonable access is defined by the WHO as a source sufficient to supply 20 liters per person per day and no further than one kilometer from the dwelling. However, water at this distance must be transported to the home and stored before drinking, increasing the risk of contamination prior to consumption. ⁸ Taking expected population growth into account, achieving the Millennium Development Goal of halving the proportion of the population without access to safe water and sanitation by 2015 (General Assembly of the United Nations 2000) in rural Africa would require providing 400 million people with improved water supplies and 200 million people with improved sanitation over the next decade (WHO 2002a).

1983, Esrey *et al.* 1985, Esrey and Habicht 1986, Esrey *et al.* 1991, Rosen and Vincent 1999, and Fewtrell *et al.* 2005). Critically, much of this research is hampered by important methodological problems, including the issues we discussed in Section 1.3.

Many studies of water infrastructure provision in developing countries lack a plausible comparison group and thus, without a credible counterfactual, cannot isolate a causal treatment effect from service provision. We are aware of two studies that rely on quasi-experimental designs (Huttly *et al.* 1987 [in Bangladesh] Aziz *et al.* 1990, Blum *et al.* 1990 [in Nigeria]), but from the published papers neither study appears to have used random assignment to treatment status. The studies are both hampered by small village-level sample sizes (both studies include only five villages in total).

To our knowledge, only one paper adequately accounts for externalities when assessing the impacts of water and sanitation investments. Watson (forthcoming) exploits the fact that a series of water and sanitation interventions introduced on Native American reservations in the United States from 1960-1998 were largely uncorrelated with other factors affecting infant health, and largely exogenous to local community characteristics after accounting for county and year fixed effects. This research suggests a 10 percent increase in the fraction of homes with improved water and sanitation service reduced Native American infant mortality by four percent. The improved disease environment reduced infant mortality rates among local residents not living on the reservation as well—a fact Watson uses as a means to measure the significant externalities associated with the program.

2.2 Sanitation service provision

The appropriate disposal of feces is another method to reduce the incidence of fecaloral disease. As in the case of drinking water, sanitation service provision is of most concern in Sub-Saharan Africa and Asia. In each of these regions, nearly one-half of the rural population does not have access to sanitation services (WHO 2000).

Relatively few evaluations of latrine provision alone have been published. As discussed in Section 1.3, sanitation service provision is usually evaluated as part of a package of interventions including water service provision and/or hygiene education.

We have identified one randomized control trial of latrine provision, which measures the relative effectiveness of latrines and insecticide spraying for trachoma control (Emerson *et al.* 2004). In this study, with a large sample and over 20 clusters of households, latrine provision did not significantly reduce trachoma incidence. Latrine provision may have affected other water-related diseases, but according to the published paper these outcomes appear not to have been measured. Several case-control studies, which compare health outcomes among children presenting at hospital or clinics with diarrhea to children with similar observable characteristics but presenting with other illnesses (such as upper respiratory infections), have found that access to latrines reduces acute diarrhea incidence (Daniels *et al.* 1990, Meddings *et al.* 2004). This research strategy is vulnerable to the same methodological critique as cross-sectional regression analyses; namely, cases that are similar across observable characteristics may differ systematically along dimensions that are difficult to measure (see Section 1.3).

Two influential papers (Esrey 1996, Esrey *et al.* 1991) are frequently cited in the literature as evidence for the relative importance of sanitation investments and hygiene

education over the provision of improved water quality (e.g. USAID 1996, Vaz and Jha 2001, World Bank 2002). In a review including the Bangladesh and Nigeria interventions discussed previously, as well as twenty-three other studies deemed by the authors to be relatively rigorous, Esrey et al. (1991) attempt to separate the relative impact of water supply, sanitation, and hygiene education interventions on diarrheal morbidity. They conclude that the median reduction in diarrheal morbidity from either sanitation supply or hygiene education provision is nearly twice the median reduction from an investment in water quality alone or an investment in water quantity and water quality together. Using multivariate regression analysis of household infrastructure status and diarrhea prevalence from several countries, Esrey (1996) reaches a similar conclusion; benefits of improved water quality occur only in the presence of improved sanitation, and only when a water source within the home is present.

We believe further research before the conclusions of Esrey and his colleagues about the relative importance of water and sanitation investments can be accepted. As a result of the observational nature of Esrey's (1996) data, these results are subject to omitted variable bias of unknown magnitude (see Section 1.3). Esrey *et al.*'s (1991) meta-analysis that reports that sanitation investments and hygiene education are more effective interventions than water quality improvements also does not convincingly resolve this research question. While the analysis focuses on what the authors deem to be rigorous studies, it appears that none of the studies has a truly randomized control trial design. Drawing conclusions about whether water quality or quantity is more important when

⁹ We also agree with Rosen and Vincent (1999) who suggest several other shortcomings of this research. The data used by Esrey includes few African households. In addition, the cross-country data cannot distinguish between working and broken taps. Finally, Esrey's conclusions may not be warranted because many of the correlations he uses to draw inferences are statistically insignificant.

wells are provided (as is the case in the most rigorous studies reviewed) is difficult because well provision simultaneously improves water quality and reduces the distance to a water source. 10 In addition, while households closer to pumps may use more water and have better health outcomes, the location of pumps is not random. Well location may be determined by where more powerful households live, for example, and this could affect health outcomes through other channels, thus confounding interpretation of results. Nor does the quasi-experimental work in Bangladesh and Nigeria convincingly show that sanitation service is more effective than the provision of new water supplies. In Bangladesh, diarrhea fell in the treatment group after the provision of boreholes but prior to the provision of latrines, without significant additional reduction after the provision of both interventions (Aziz et al. 1990). In Nigeria, delays in the latrine construction component of the project lead the authors to conclude the weight-for-height improvements they observed largely reflect the effects of well construction (Huttly et al. 1987). Finally, Fewtrell et al. (2005) present a more recent meta-analysis that finds that water quality improvements provided via point-of use technologies may be at least as effective as sanitation and hygiene improvements. This evidence is based on rigorous randomized control trials, as we discuss further below. However, this recent review confirms the continuing dearth of credible evidence on the effectiveness and costeffectiveness of hardware investments.¹¹

¹⁰ This is equally true for other studies that examine the impact of borehole or tap construction (Mason *et al.* 1986, Linkskog *et al.* 1987, El Kholy *et al.* 1989).

¹¹ While it is difficult to draw conclusions about the benefits of hardware investments to improve water quality improvements from existing micro-epidemiological studies, other evidence suggests that the benefits are very large. Cutler and Miller (2005) use exogenous variation in the timing and location of water filtration and chlorination technology adoption across U.S. cities in the 19th Century to identify the contribution of improved water quality to the epidemiological transition in American cities. They find that clean water was responsible for about one-half of the observed mortality decline in cities, and nearly two-thirds of the child mortality reduction.

2.3 The maintenance of water and sanitation infrastructure investments

Water and sanitation infrastructure require management and upkeep in order to realize the benefits of these investments over a long time horizon. In this section, we review the available evidence on various infrastructure management schemes, cost-sharing efforts, and on the involvement of women in the management of public good provision.

Infrastructure maintenance has historically been a major problem in developing countries and in the rural water sector in particular. For instance, a quarter of India's water infrastructure is believed to be in need of repair (Ray 2004). The *World Development Report* (World Bank 2004b) estimates that more than one-third of existing rural water infrastructure in South Asia is not functional. Miguel and Gugerty (forthcoming) report that in western Kenya, nearly 50 percent of borehole wells dug in the 1980s, and subsequently maintained using a community-based maintenance model, had fallen into disrepair by 2000. Difficulties with maintaining water infrastructure, particularly in rural areas, reduce the cost-effectiveness of these interventions relative to other measures that prevent diarrhea. The challenge of identifying institutional structures that will make such investments financially sustainable is an important area of ongoing research.

2.3.1 Community-level infrastructure management schemes

A standard model for maintaining donor-funded infrastructure projects in developing countries is to establish user groups responsible for project maintenance and management. This institutional design is expected to result in improved financial sustainability. Giving communities direct control or ownership over key project decisions

is intended to improve the quality of public services, while reducing the need for ongoing donor funding or involvement.¹²

Despite donor interest in community-based development schemes, there is little convincing empirical evidence that local user-committee management of local public goods such as improved drinking water sources results in either greater financial sustainability or better quality service, relative to on-going centralized public funding. In a recent comprehensive review of the literature on community-based development projects, Mansuri and Rao (2004) note that existing research examining "successful" community-based projects does not compare these projects to centralized mechanisms for service delivery or infrastructure maintenance. This makes it difficult to determine whether alternative project designs would have had different results.¹³ The limited available empirical evidence suggests the impact of the community-based development approach on infrastructure maintenance is mixed at best.¹⁴

A potential alternative to community-managed local public goods that deserves more research is the cost-effectiveness of technologies that allow households or small groups (either newly-organized or pre-existing groups) of households to self-supply water services. For example, pumps that provide both drinking and irrigation water to a few

¹² Other goals may also be important, as community-based development may be expected to empower poor people or strengthen local governance.

There may well be a strong interaction between the technical difficulty of maintaining a particular infrastructure and the functioning of community based maintenance groups. For example, the task of maintaining a spring is very different from maintaining a hand pump. Keeping springs clean may require many small actions by a number of people in a community, while maintaining a hand pump often requires financing for spare parts, technical knowledge to make repairs, and special equipment to lift the pump out of the ground before it can be repaired.

¹⁴ Isham and Kahkonen (1999) find that in their study of water projects in 44 Indonesian villages "the existence of [local] water committees had either no effect or a negative effect on service performance," though greater community participation in the design of community-based water projects did improve water supply and health outcomes somewhat. On the other hand, Khwaja (2003) finds that community-managed projects in Pakistan, including irrigation and drinking water investments, performed better than projects implemented by the government without community participation. However, he also finds that project-specific factors, such as the quality of the outside facilitator, may have a larger impact on project success than community characteristics.

households may provide sufficient private income benefits such that free-riding problems can be overcome. Randomized impact evaluations that compare the cost-effectiveness of self-supply schemes to infrastructure services that serve a larger community are needed.

2.3.2 Cost-sharing

Similar to efforts to promote community-based management of infrastructure in rural areas of developing countries, cost-sharing with user fees has been widely advocated as necessary for the sustainability of public health services in many less developed countries. However, it remains unclear precisely how user fees have affected the utilization of health services in many countries (McPake 1993; Gertler and van der Gaag 1990). Several studies provide suggestive evidence that continued support (financial and otherwise) may be necessary for effective infrastructure investments and public service delivery (Katz and Sara 1998, Kleemeier 2000, Dayton-Johnson 2000, Newman *et al.* 2002, Kremer and Miguel 2004). Certainly, further research on this question is needed that transparently compares the counterfactual of subsidized public service provision to community-based management schemes.

2.3.3 Involving women in the management of environmental public goods

Some of the sociology literature has predicted that increasing the involvement of women on user committees will improve the management of collectively-owned natural resources because women's social networks provide prior experience with collective action (Agarawal 2000). In addition, since women are major users of these goods, their involvement with the creation and thus their compliance with rules may be especially important (Zwarteveen and Meinzen-Dick 2001).

However, the evidence to date on the impact of women's involvement in public good management is limited. Efforts using laboratory experiments to assess whether women supply differing levels of public goods or are more cooperative than men (Nowell and Tinkler 1994, Eckel and Grossman 1998, Solow and Kirkwood 2002) seem to be sensitive to the form of the experiment performed. Much of the empirical field evidence on this question is hampered by concerns about reverse causality (*e.g.* Dollar *et al.* 2001). It is difficult to determine whether the inclusion of women causes a particular outcome to occur, whether the fact that an outcome occurs encourages the participation and inclusion of women, or whether some other factors are driving these results. This problem affects both retrospective analyses (*e.g.* Prokopy 2004) and case studies (INSTAW-UN 1990, Wijk-Sijbesma 1998, Gross *et al.* 2001), neither of which is able to establish a causal relationship between women's participation and observed outcomes.

Chattopadhyay and Duflo (2004) are able to identify a causal relationship between women's participation and project outcomes in their study of a randomized policy change in India that increased the role of women in policy decision-making. They show that village councils headed by women were more likely to invest in public infrastructure for drinking water, and, more generally, that councils dominated by a particular gender were more likely to invest in goods important to that gender. These results suggest that, if women tend to value water resources more than men, placing them in charge of environmental public health infrastructure may be a cost-effective policy for donors and governments.¹⁵ Additional evidence on these questions in other settings is needed.

¹⁵ Of course, the composition of user committees may vary in ways that are not as easily observable as gender composition, but that are important for determining the success of coordination and collective action for public goods management. Theories of social capital (Putnam 1993, Coleman 1990) argue that the trust, norms,

3 Individual-level behavior change

Concerns about the effectiveness of source water quality improvements have led to research exploring several types of individual-level investments that could reduce the diarrheal disease burden. Point-of-use water treatment systems (3.1) are intended to reduce pathogen loads resulting from environmental exposure prior to consumption, rather than at the source. Drawing on research contending that if communities are to benefit from improved water and sanitation technology they must understand the causes of diarrheal diseases and the behaviors facilitating their transmission, health education/promotion (3.2) and hand-washing promotion (3.3) "software" efforts have also been studied. We review the available evidence on these hygiene behavior modifications, and discuss (3.4) the need for further work investigating how these interventions may be implemented.

3.1 Point-of-use water treatment

A key reason to be concerned about the cost effectiveness of water supply improvements at the source is the possibility of water contamination during storage and transportation (Esrey *et al.* 1991, Wright *et al.* 2004). Point-of use water treatment can address these issues. These systems include storage containers with narrow mouths, secure lids, and dispensing spigots, which eliminate contact between hands and stored water and help to prevent contamination, as well as chemical disinfectants, which may be simple household bleach or other products such as flocculent disinfectants. These chemical products kill pathogenic bacteria when added to microbacteriologically-

and networks created by participation in social organizations can facilitate collective action in local communities.

contaminated water; however, although these products may make the water cleaner they may also negatively affect the taste of the water.

Evidence from randomized evaluations assessing the health impacts of interventions to improve water quality at the point of use suggests this may be a promising way to reduce diarrheal morbidity (Semenza *et al.* 1998, Quick *et al.* 1999, Quick *et al.* 2002, Sobsey *et al.* 2003, Reller *et al.* 2003, Clasen *et al.* 2004). Randomized impact evaluations of point-of-use water treatment systems suggest that, with a rate of technology uptake on the order of 70 percent (over the very short run and with frequent contact with field workers), large statistically significant reductions in diarrheal incidence on the order of 20 to 30 percent are observed at the household level even after taking intra-household correlation into account (Quick *et al.* 1999, Reller *et al.* 2003). ¹⁶ ¹⁷

The next step will be to examine the cost-effectiveness and sustainability of these interventions. Little is known about whether people will be willing to permanently adopt systems that affect the taste of water, slow the rate at which water can be consumed (for example, due to filtration), or require that traditional storage containers be abandoned.

¹⁶ Despite these large reported gains however, many of these same studies also find the observed reductions in diarrheal morbidity associated with the intervention are concentrated among children under age one, and among children over age five. Surprisingly, the age group with the highest diarrheal morbidity that is, children between ages one and five, may be least affected by this intervention (Quick et al. 1999, Reller et al. 2003, Sobsey et al. 2003). However, in two cases larger gains from point-of-use water treatment are identified in children under age five. In Uzbekistan, diarrhea incidence in children under five fell by 85 percent after the provision of a chlorine-stock solution and a narrow-necked storage container. Clasen et al. (2004) provided hygiene education to households that received the treatment systems, but argue this limited education effort was superfluous as households in this area of Uzbekistan already had uniformly high levels of hygiene practices. In rural Bolivia Clasen et al. (2004) identified reductions in diarrhea incidence of about 83 percent in children under age five with a sample size of 30 children in 50 households. In this case, the treatment system provided to households was a ceramic water filter that left the taste of water unchanged, rather than a disinfectant and a container as in the Uzbekistan study. Uptake appears to have been on the order of 70 percent, suggesting that improved compliance with treatment relative to the studies that supplied disinfectant cannot explain the observed result. It is unclear why the large gains identified in these two studies for the most vulnerable group have not been found in other contexts.

¹⁷ A few randomized trial studies report evidence on solar disinfection of water in homes that support these research findings by identifying large benefits for young children. In a Kenyan Maasai community researchers identified large health benefits for children under age six from the exposure of drinking water to sunlight (Conroy *et al.* 1999, 2001).

Willingness to pay for these systems has not been investigated on any meaningful scale using a randomized approach. Household willingness to use these treatment systems on a consistent basis, and ultimately to pay for them, remains to be verified (Makutsa *et al.* 2001). As we discuss in Section 5, research investigating these issues is a critical open area for future work. Research is also needed to compare the impacts of this intervention to investments that improve source water quality alone.

3.2 Health education and health promotion

The literature evaluating health education programs¹⁸ in developing countries suffers from important methodological weaknesses similar to those that discussed previously in the context of water and sanitation interventions.¹⁹ Loevinsohn (1990) and Cave and Curtis (1999) survey this research and conclude that methodological improvements are needed (see also Fewtrell et al. [2005]). Cave and Curtis (1999) are able to identify only five studies since 1987 in which sample sizes exceeded 60 individuals or two clusters and in which randomized or quasi-randomized designs were used (Haggerty *et al.* 1994, Lloyd *et al.* 1994, Pant *et al.* 1996, Tayeh *et al.* 1996, and Elder *et al.* 1998).

Among studies that do use a randomized design, several have found positive effects of health education. For example, Haggerty *et al.* (1994) find an 11 percent reduction in the risk of reporting diarrhea as a result of an education program randomly administered to a sample of 2,000 children from 18 communities in rural Zaire (now Democratic

¹⁸ Health education programs may target one particular hygiene behavior, or a set of inter-related behaviors. One well-known education program is the Participatory Hygiene and Sanitation Transformation Series (PHAST) education methodology. This widely-used program, developed by WHO, United Nations Development Program, and the World Bank Water and Sanitation Program, emphasizes increasing awareness of the causes of diarrhea, the reasons that diarrhea is dangerous, and strategies for treatment and prevention (Wood *et al.* 1998).

As an additional complication, measuring behavior change can be both costly and difficult. For example, in Congo, Manun'Ebo et al. (1997) investigate the relationship between reported feces-handling behavior and observed feces-handling behavior and find that agreement was little better than might be expected by chance.

Republic of the Congo). In urban Bangladesh, Stanton and Clemens (1987) identify a 25 percent reduction in diarrhea incidence six months after a hygiene education program was randomly administered to 1,900 families in 51 communities.

However, the relative cost-effectiveness of education relative to other interventions remains uncertain and may differ systematically across groups. Pant *et al.* (1996) divide a random sample of 40,000 children from 75 locations in Nepal into two randomly-assigned groups and provide one group with Vitamin A supplements and the other with nutrition education. They find that the risk of child mortality declined by equal amounts in the two groups but that the education program was more costly to deliver. In addition, they also find that education was the least cost-effective when maternal literacy was low. This study points to the need for further work that directly compares alternative interventions to each other.

3.3 Handwashing

Hands are a common vector for the transmission of fecal-oral diseases (Curtis *et al.* 2000). Hand-washing with soap after defecation, after cleaning children, and before and after cooking can block this diarrheal disease transmission path.

A recent survey of the literature (Curtis and Cairncross 2003) evaluating the effect of hand-washing with soap on diarrhea risk in developing countries (and in daycare settings in developed countries) identified only two randomized control trials with more than two communities in their samples (Khan 1982, Han and Hlaing 1989).²⁰ Each of

²⁰ Low hand-washing rates in U.S. elementary schools have prompted interest in the use of instant hand sanitizers (which do not require water) in classrooms as an alternative means of breaking disease transmission. A review of the evidence on the efficacy of sanitizers (Meadows and LeSaux 2004) identified only one (clustered) randomized control trial of this technology to date. That trial found that sanitizers can effectively

these studies have relatively large sample sizes randomly divided into treatment and control groups and measure compliance by observing or weighing provided bars of soap as well as by tracking diarrhea cases. We also identified an additional recent randomized impact evaluation of a hand-washing promotion program in Karachi, Pakistan (Luby *et al.* 2004a, 2004b).

Each of these randomized impact evaluations reports large positive effects of hand washing and soap provision programs on diarrhea incidence. Khan (1982) reports that either the provision of soap and water storage containers or the provision of soap alone, along with an initial exhortation to increase the frequency of hand washing, reduced individual-level shigella reinfection by 67 percent in Bangladesh. However, this result does not account for intra-household clustering, thus overestimating the statistical significance of the effect of the intervention. Han and Hlaing (1989) report a 40 percent reduction in dysentery incidence among children under age two (though no reduction in incidence for older children) following hand washing education and the provision of soap to a random sample of mothers in Rangoon (Yangon). Luby et al. (2004a) report similar results. In their study, infants and malnourished children under age five living in treatment households had 39 percent fewer days of diarrhea compared to the control group after one year of treatment and observation. Gains roughly doubled after a second year of service provision as relatively poor households increased their compliance (Luby et al. 2004b).

reduce absenteeism as a result of illness (White et al. 2001). Based solely on that study, it is unclear whether sanitizers may be suitable for a developing country context.

3.4 The implementation of "software" interventions

The effectiveness of various software interventions as actually implemented in practice and over time is a critical determinant of the relative cost-effectiveness of each intervention. The health benefits of this class of interventions rely on individual-level decisions to adopt and consistently adhere to certain behaviors. The studies reviewed in this section suggest several unresolved questions about the efficacy of individual-level behavior change efforts in practice.

First, the cost-effectiveness of handwashing and soap provision programs is uncertain. Weekly or daily reminders, which Luby *et al.* (2004b) identify as critical for adoption by the poorest households as compared to richer households (and upon which all of these studies rely), are prohibitively expensive to provide on a large-scale basis. The cost-effectiveness of point-of-use water treatment systems also needs further study; to our knowledge no evaluation has been published that compares health gains in groups that have been charged different amounts (or no fee) for treatment systems.

Second, existing research has not established that technology adoption persists in the long term. For instance, Khan (1982) tracks his subjects for only 10 days, and the other studies, while longer-term, rely on daily or weekly visits for observational data and reinforcement of treatment. We have little evidence from this research that hand-washing behaviors will persist after fieldworkers are gone, and evidence on the persistence of other hygiene behavior modifications is mixed. Some follow-up studies taking place several years after soap provision and hygiene behavior change efforts find that the behavior change can persist (Wilson and Chandler 1993, Shordt and Cairncross in press). However, as the initial interventions studied were not randomized, these finding is

somewhat difficult to interpret. On the other hand, a follow-up study performed five years after the quasi-randomized intervention in Bangladesh that we described in section 2.1 suggests that the hygiene education effort there may have been relatively ineffective as five years later treatment households did not exhibit better hygiene behaviors than control households (Hoque *et al.* 1996).

In order to identify cost-effective means of facilitating long-term behavior change and technology adoption, additional research is needed that compares alternative messages and alternative message delivery avenues in several cultural contexts. For example, randomized evaluations are needed that compare health education messages directed towards women that emphasize family health versus children's well-being in particular. The usefulness of positive and negative messages should also be explored.

In some contexts, and for some behaviors, households may not be the most costeffective entry point for message dissemination. Evaluations are needed that examine the
relative effectiveness and cost-effectiveness community-level hygiene education efforts
(along the lines of PHAST) that target women with other designs that might include
programs that work through maternity clinics (potentially providing incentives for
women to take-up antenatal services) or school-based programs. It is possible, for
example, that providing soap, hand washing facilities, and messages about the importance
of hand washing after defecation in primary schools may be an effective and relatively
cost-effective means of promoting long-term behavior change.

Research that identifies how social learning occurs about hygiene behavior and water treatment is also needed. For example, we know little about whom opinion leaders are for water-related matters, and identifying these individuals and encouraging their

adoption of treatment systems or hygiene behaviors may be critical to encouraging wider community adoption. Research might examine whether encouraging women's groups to use water treatment systems is particularly effective as a means of increasing uptake, for example, as this could be a means of targeting mothers of young children. Such targeted dissemination to women could be compared to targeted dissemination to political leaders (likely men) and wider service provision.

Service pricing decisions may have implications for both short-run cost effectiveness calculations and long-term adoption rates. The implications of cost-recovery efforts for uptake of point-of-use treatment systems in particular is little understood. Social learning effects may make it desirable to offer technologies for free initially to at least some populations. On the other hand, some argue that offering a service free of charge, even temporarily, may make people reluctant to pay for it later. The effects of pricing on service use and uptake needs further study.

Further research on the use of social marketing to increase or maintain adoption rates in this sector also would be fruitful. Social marketing can substitute for or complement educational programs, and applies marketing techniques to design and implement programs promoting socially beneficial behavior change. This approach is growing in both popularity and use within the public health community and elsewhere (Grier and Bryant 2005).

4 Conclusion and recommendations

Randomized control trials have established that vaccination, ORT, breastfeeding, and micronutrient supplementation are effective in reducing the burden of diarrhea, and that these are cost-effective investments. This evidence base has led to widespread

implementation of these interventions, in turn saving millions of lives. The continuing toll of diarrhea makes it clear that further investments to prevent this disease are needed. Yet we have scant evidence and only tenuous consensus on the impact and cost-effectiveness of various environmental health interventions that might complement existing child health programs.

We have tentatively identified three broad areas for future work to address this shortcoming of the existing literature: First, research is needed that evaluates alternative environmental health interventions on a comparable basis across several settings. By properly assessing the life of infrastructure investments and accounting for externalities, researchers' goal should be to determine the extent to which environmental health projects can cost-effectively complement the medical interventions currently included in child health programs. Work is needed to rigorously assess the relative effectiveness of investments in improving water quality and increasing water quantity and to determine the extent to which contamination in water transport and hygiene practices may reduce the benefits of water quality improvements at the source. This will allow researchers to trace the chain of transmission from water source to health outcomes more accurately than is currently possible.

Second, future studies should expand efforts to understand the determinants of individual level technology adoption (including social learning) and increase technology uptake in this context. Additional information is needed about households' willingness to permanently adopt point-of-use water treatment systems, and to modify hygiene behaviors on a consistent basis. The implications of cost-recovery efforts, alternative message designs, and alternative means of product introduction for adoption should be

explored. Follow-up studies should be planned when initial evaluations are designed to determine whether behavior change persists over the longer term.

Third, future research should also expand efforts to assess the quality of infrastructure maintenance under different management schemes, again using a randomized impact evaluation methodology. In particular, research is needed that compares community-based maintenance schemes with centralized mechanisms of service delivery, and that estimates which approach is the most cost-effective way to achieve financially sustainable rural water and sanitation supply. A recent *World Development Report* (World Bank 2004b) cites local governments, regional utilities, and independent providers, in addition to local community groups, as possible institutional channels for supporting community-based water systems. Research designs might directly compare the success of several of these models in infrastructure maintenance.

Several key methodological points should also be integrated into future work. First, to the greatest extent possible, projects should be designed as randomized evaluations.

Randomized evaluations allow researchers to credibly estimate the impacts a particular intervention, unlike many existing non-experimental studies that are likely to suffer from omitted variable bias. By comparing results from randomized evaluations to those that would be found in cross-sectional regression, researchers will also be able to identify and better understand the biases in cross-sectional regressions. Prospective randomized evaluations also allow for the collection of the data necessary to derive cost-effectiveness estimates. A fruitful area for future work would be to examine the longer-term impacts of some of the earlier, well-executed randomized trials discussed in this review.

Second, great effort should be made to significantly expand sample sizes beyond those seen in most existing studies. Larger sample sizes allow researchers to take intracluster correlation into account in statistical analysis and to detect moderate treatment effects with a reasonable degree of power.

Finally, projects should be designed so as to allow for the estimation of the impact of individual interventions, rather than to only estimate the impact of packages of interventions. For example, there is a need for credible information on the complementarities and possibilities for substitution between hardware and software interventions. Evidence is also needed comparing the impacts of water quality improvements across households with different levels of sanitation, thus allowing researchers to rigorously assess the claim that improved water quality is only effective after sanitation has been improved.

References

- Agarwal, B. (2000), "Conceptualizing environmental collective action: Why gender matters," Cambridge Journal of Economics 24, 283-310.
- Angrist, J. *et al.* (1996), "Identification of causal effects using instrumental variables," Journal of the American Statistical Association 91(434), 444-472.
- Aziz, K. *et al.* (1990), "Reduction in diarrhoeal diseases in children in rural Bangladesh by environmental and behavioral modifications," Transactions of the Royal Society of Tropical Medicine and Hygiene 84(3), 433-438.
- Beaton, G.H. *et al.* (1993), "Effectiveness of vitamin A supplementation in the control of young child mortality in developing countries," Nutrition Policy Discussion Paper number 13 ACC/SCN: Geneva.
- Berkman, D.S. *et al.* (2002), "Effects of stunting, diarrhoeal disease, and parasitic infection during infancy on cognition in late childhood: A follow-up study," Lancet 359, 564-71.
- Black, R.E. (1998), "Therapeutic and preventive effects of zinc on serious childhood infections diseases in developing countries," American Journal of Clinical Nutrition 68(supplement), 476s-9s.
- Blum, D. *et al.* (1990), "The Imo State (Nigeria) drinking water supply and sanitation project, 2. Impact on dracunculiasis, diarrhoea and nutritional status," Transactions of the Royal Society of Tropical Medicine and Hygiene 84(2), 316-321.
- Blum, D. and R.G. Feacham (1983), "Measuring the impact of water supply and sanitation investments on diarrheal diseases: Problems of methodology," International Journal of Epidemiology 12(3), 357-65.
- Briend, A. (1990), "Is diarrhea a major cause of malnutrition among the under-fives in developing countries? A review of available evidence," European Journal of Clinical Nutrition 44, 611-28.
- Briscoe, J. (1984), "Water supply and health in developing countries: Selective primary health care revisited," American Journal of Public Health 74(9), 1009-1013.
- Cairncross, S. (1996), Water quality, quantity and health, in J.O. Drangert, *et al.* eds., "Conference on safe water environments," Water and Environmental Health Studies No. 24. Linkoping University, Sweden.
- Cave, B. and V. Curtis (1999), "Effectiveness of promotional techniques in environmental health," London School of Hygiene and Tropical Medicine WELL Study Task # 165.

- Chattopadhyay, R., and E. Duflo (2004), "Women as policy makers: Evidence from a randomized policy experiment in India," Econometrica 72(5), 1409-1443.
- Chavez, A.V. and C. Martinez (1986), "Efectos de la desnutrición moderada sobre el rendimiento escolar temprano," La Revista de Investigación Clinica 38 (Suplemento), 115-19.
- Child Health Research Project (1998), "Childhood diarrhea in Sub-Saharan Africa," Child Health Research Project Special Report 2(1).
- Clasen, T.F. *et al.* (2004), "Reducing diarrhea through the use of household-based ceramic water filters: A randomized controlled trial in rural Bolivia," American Journal of Tropical Medicine and Hygiene 70(6), 651-57.
- Coleman, J. (1990), Foundations of social theory, Cambridge: Harvard University Press.
- Conroy, R.M. *et al.* (1999), "Solar disinfection of water reduces diarrheal disease: An update," Archives of Disease in Childhood 81, 337-8.
- Conroy, R.M. *et al.* (2001), "Solar disinfection of drinking water protects against cholera in children under six years of age," Archives of Disease in Childhood 85, 293-5.
- Curtis, V. and S. Cairncross (2003), "Effect of washing hands with soap on diarrhea risk: A systematic review," Lancet Infectious Disease 3(5), 275-281.
- Curtis, V. *et al.* (2000), "Domestic hygiene and diarrhea- pinpointing the problem," Tropical Medicine and International Health 5(1), 22-32.
- Cutler, D. and G. Miller (2005), "The Role of public health improvements in health advances: The 20th Century United States," National Bureau of Economic Research (NBER) Working Paper # 10511.
- Daniels, D.L. *et al.* (1990), "A case-control study of the impact of improved sanitation on diarrhea morbidity in Lesotho," Bulletin of the World Health Organization 68(4), 455-464.
- Dayton-Johnson, J. (2000), "The determinants of collective action on the commons: A model with evidence from Mexico," Journal of Development Economics 62, 181-208.
- Dollar, D. *et al.* (2001), "Are women really the 'fairer' sex? Corruption and women in government," Journal of Economic Behavior and Organization 46(4), 423-29.
- Eckel, C. and P.J. Grossman (1998), "Are women less selfish than men? Evidence from dictator experiments," Economic Journal 198, 726-35.

- el Kholy, H. *et al.* (1989), "Effects of borehole wells on water utilization in Schistosoma haematobium endemic communities in Coast Province, Kenya," American Journal of Tropical Medicine and Hygiene 41(2), 212-19.
- Elder, J.P. *et al.* (1998), "Project Salsa: development and institutionalization of a nutritional health promotion project in a Latino community," American Journal of Health Promotion 12(6), 391-401.
- Emerson, P.M. *et al.* (2004), "Role of flies and provision of latrines in trachoma control: Cluster-randomized controlled trial," Lancet 363, 1093-1104.
- Esrey S.A. (1996), "Waste, water and well-being: A multicountry study," American Journal of Epidemiology 143(6), 608-22.
- Esrey S. A. *et al.* (1985), "Interventions for the control of diarrheal diseases among young children: Improving water supplies and excreta disposal facilities," Bulletin of the World Health Organization 63(4), 757-72.
- Esrey S. A. and J.-P. Habicht (1986), "Epidemiologic evidence for health benefits from improved water and sanitation in developing countries," Epidemiology Review 8, 117-128.
- Esrey S.A. *et al.* (1991), "Effects of improved water supply and sanitation on ascariasis, diarrhea, dracunculiasis, hookworm infection, schistosomiasis and trachoma," Bulletin of the World Health Organization 60, 609-621.
- Feacham, R.A. (1977), Water supplies for low-income communities: Resource allocation, planning and design for a crisis situation, in R. Feacham, M. McGarry, and D. Mara eds., "Waste, water, and health in hot climates," John Wiley and Sons: London.
- Fewtrell, L. *et al.* (2005), "Water, sanitation, and hygiene interventions to reduce diarrhoea in less developed countries: A systematic review and meta-analysis," Lancet Infectious Diseases 5, 42-52.
- Galiani, S. *et al.* 2002, Water for life: The impact of the privatization of water services on child mortality, Center for Research on Economic Development and Policy Reform, WP 154, August.
- General Assembly of the United Nations (2000), United Nations Millennium Declaration, Resolution adopted by the General Assembly 55/2.
- Gertler, P. and J. van der Gaag. (1990), The willingness of pay for medical care: evidence from two developing countries. Johns Hopkins University Press: Baltimore.

- Glass, R. *et al.* (2004), Rotavirus vaccines, in C.A. de Quadros, ed., "Vaccines: Preventing Disease & Protecting Health," Pan American Health Organization: Washington, D.C.
- Grier, S. and C. Bryant (2005), "Social marketing in public health," Annual Review of Public Health 26, 319-339.
- Gross, B. *et al.* (2001), Linking sustainability with demand, gender and poverty: A study in community-managed water supply projects in 15 countries, Washington, D.C.: World Bank Water and Sanitation Program.
- Grotto, I. et al. (2003), "Vitamin A supplementation and childhood morbidity from diarrhea and respiratory infections: A meta-analysis," Journal of Pediatrics 142, 297-304.
- Guerrant, D.I. *et al.* (1999), "Association of early childhood diarrhea and cryptosporidiosis with impaired physical fitness and cognitive function four-seven years later in a poor urban community in northeast Brazil." American Journal of Tropical Medicine and Hygiene 61(5), 707-13.
- Guerrant, R.L. *et al.* (1992), "Diarrhea as a cause and effect of malnutrition: Diarrhea prevents catch-up growth and malnutrition increases diarrhea frequency and duration," American Journal of Tropical Medicine and Hygiene 47 (1 part 2), 28-35.
- Haggerty P.A. *et al.* (1994), "Community-based hygiene education to reduce diarrhoeal disease in rural Zaire: impact of the intervention on diarrhoeal morbidity," International Journal of Epidemiology 23, 1050-1059.
- Han, A. M. and T. Hlaing (1989), "Prevention of diarrhoea and dysentery by hand washing," Transactions of the Royal Society of Tropical Medicine and Hygiene 83(1), 2128-31.
- Hill, Z. *et al.* (2004), Family and community practices that promote child survival, growth, and development: A review of the evidence, WHO: Geneva.
- Hoque, B.A. *et al.* (1996), "Sustainability of a water, sanitation and hygiene education project in rural Bangladesh: A 5-year follow-up," Bulletin of the World Health Organization 74(4), 431-7.
- Huttly S.R.A. *et al.* (1987), "The epidemiology of acute diarrhea in a rural community in Imo State, Nigeria," Transactions of the Royal Society of Tropical Medicine and Hygiene 81, 865-70.
- Hutton, G. (2001), Economic evaluation and priority setting in water and sanitation interventions, in L. Fewtrell and J. Bartam (Eds.), "Guidelines, standards and health: Assessment of risk and risk management for water-related infectious disease," WHO: Geneva.

INSTAW-UN, (1990), "Women, water and sanitation," in S. Sontheimer (ed.), Women and the environment: A reader, New York, NY: Monthly Review Press, 119-132.

International Institute for Environment and Development (2000), Drawers of Water II: Thirty years of change in domestic water use and environmental health in East Africa, International Institute for Environment and Development: London.

Isham, J. and S. Kahkonen (1999), "What determines the effectiveness of community-based water projects? Evidence from central Java, Indonesia on demand responsiveness, service rules, and social capital," Social Capital Initiative Paper 14, World Bank, Environmentally and Socially Sustainable Development Network, Washington, DC.

Katz, T. and J. Sara (1998), "Making rural water supply sustainable: Recommendations from a global study." UNDP-World Bank Water and Sanitation Program. Available At: http://www.wsp.org/publications/global_ruralstudy.pdf.

Khan, M.U. (1982), "Interruption of shigellosis by hand washing," Transactions of the Royal Society of Tropical Medicine and Hygiene 76(2), 164-8.

Khwaja, A.I. (2003), "Can good projects succeed in bad communities? The determinants of collective success in maintaining local public goods," Harvard Kennedy School of Government Working Paper RWP01-043.

Kleemeier, E. (2000), "The impact of participation on sustainability: an analysis of the Malawi rural piped scheme program," World Development 28(5), 929-944.

Kosek, M. *et al.* (2003), "The global burden of diarrheal disease, as estimated from studies published between 1992 and 2000," Bulletin of the World Health Organization 81, 197-204.

Kremer, M. and E. Miguel (2004), "The illusion of sustainability," National Bureau of Economic Research (NBER) Working Paper #10324.

Lancet (1991), "Diarrhea and malnutrition," Lancet 338, 921-22.

Lenehan, A. and J. Martin (1997), "Spring protection in southern KwaZulu Natal," Paper presented at the twenty-third Water, Engineering, and Development Centre (WEDC) Conference on Affordable Water Supply and Sanitation, Durban.

Lindskog, U. *et al.* (1987), "Child health and household water supply: A longitudinal study of growth and its environmental determinants in rural Malawi," Human Nutrition, Clinical Nutrition 41(C), 409-423.

- Lloyd, L.S. *et al.* (1994), "The design of a community-based health education intervention for the control of Aedes aegypti," American Journal of Tropical Medicine and Hygiene 50(4), 401-11.
- Loevinsohn, B.P. (1990), "Health education interventions in developing countries: A methodological review of published articles," International Journal of Epidemiology 19(4), 788-94.
- Luby, S. *et al.* (2004a), "A low-cost intervention for cleaner drinking water in Karachi, Pakistan." International Journal of Infectious Diseases 5(3), 144-50.
- Luby, S. *et al.* (2004b), "Effect of intensive hand washing promotion on childhood diarrhea in high-risk communities in Pakistan: A randomized control trial." Journal of the American Medical Association 291(21), 2547-54.
- Mansuri, G. and V. Rao (2004), "Community based (and driven) development: A critical review," World Bank Research Observer 19(1): 1-39.
- Manun'Ebo, M. *et al.* (1997), "Measuring hygiene practices: A comparison of questionnaires with direct observations in rural Zaire," Tropical Medicine and International Health 2(11), 1015-21.
- Martines, J. *et al.*, eds. (1993) "Disease Control Priorities in Developing Countries," Oxford University Press: Oxford, U.K. and New York, p. 91.
- Martorell, R. (1993), "Enhancing human potential in Guatemalan adults through improved nutrition in early childhood," Nutrition Today, 6-13.
- Mason, P.R. *et al.* (1986), "Piped water supply and intestinal parasitism in Zimbabwean schoolchildren," Transactions of the Royal Society of Tropical Medicine and Hygiene 80(1), 88-93.
- McPake, B. (1993), "User charges for health services in developing countries: A review of the economic literature," Social Science and Medicine 36(11), 1397-1405.
- Meadows, E. and N. Le Saux (2004), "A systematic review of the effectiveness of antimicrobial rinse-free hand sanitizers for prevention of illness-related absenteeism in elementary school children," BMC Public Health 4(50).
- Meddings, D.R. *et al.* (2004), "Cost effectiveness of a latrine revision programme in Kabul, Afganistan," Bulletin of the World Health Organization 82(4), 281-289.
- Miguel E., and M.K. Gugerty (forthcoming), "Ethnic divisions, social sanctions, and public goods in Kenya," Journal of Public Economics.

- Miguel, E. and M. Kremer (2004), "Worms: Identifying impacts on education and health in the presence of treatment externalities," Econometrica 72(1), 159-217.
- Miller, P. and N. Hirschhorn (1995), "The effect of a national control of diarrheal diseases program on mortality: The case of Egypt," Social Science Medicine 40(10), 1-30.
- Mwami, J. (1995), "Spring protection- sustainable water supply," Paper presented at the twenty-first Water, Engineering, and Development Centre (WEDC) Conference on Affordable Water Supply and Sanitation, Kampala.
- Newman, J. *et al.* (2002), "An impact evaluation of education, health, and water supply investments by the Bolivian Social Investment Fund," World Bank Economic Review 16, 241-274.
- Niehaus, M.D. *et al.* (2002), "Early childhood diarrhea is associated with diminished cognitive function four to seven years later in children in a northeast Brazilian shantytown," American Journal of Tropical Medicine and Hygiene 66(5), 590-3.
- Nowell, C. and S. Tinkler (1994), "The influence of gender on the provision of a public good," Journal of Economic Behavior and Organization 25, 25-36.
- Okun, D.A. (1988), "The value of water supply and sanitation in development: An assessment," American Journal of Public Health 78(11), 1463-1467.
- Pant C.R., *et al.* (1996), "Impact of nutrition education and mega-dose vitamin A supplementation on the health of children in Nepal," Bulletin of the World Health Organization 74(5), 533-45.
- Pelltier, D. *et al.* (1995), "The effects of malnutrition on child mortality in developing countries," Bulletin of the World Health Organization 73(4), 443-8.
- Perera, B.J.C. *et al.* (1999), "The impact of breastfeeding practices on respiratory and diarrhoeal disease in infancy: A study from Sri Lanka," Journal of Tropical Pediatrics 45, 115-8.
- Pollitt, E. *et al.* (1993), "Early supplementary feeding and cognition," Monographs of the Society for Research in Child Development Serial #235, 58(7).
- Prokopy, L.S. (2004), "Women's participation in rural water supply projects in India: Is it moving beyond tokenism and does it matter?" Water Policy 6(2), 103-116.
- Putnam, R. D. (1993), Making democracy work: Civic traditions in modern Italy. Princeton: Princeton University Press.

- Quick, R.E., *et al.* (1999), "Diarrhea prevention in Bolivia through point-of-use disinfection and safe storage: A promising new strategy." Epidemiology Infections 222, 83-90.
- Quick, R.E. *et al.* (2002), "Diarrhea prevention through household-level water disinfection and safe storage in Zambia." American Journal of Tropical Medicine and Hygiene 66(5), 584-589.
- Ramakrishnan, U. and R. Matorell (1998), "The role of vitamin A in reducing child mortality and morbidity and improving growth," Salud Publica de Mexico 40(2), 189-198.
- Ray, I. (2004), "Water for all? Peri-urban and rural water delivery options: The case of India." Presentation: UC Berkeley Energy and Resources Group Fall Colloquium Series, 6 October.
- Raisler, J. *et al.* (1999), "Breastfeeding and infant illness: A dose-response relationship?" American Journal of Public Health 89, 25-30.
- Reller, M.E., *et al.* (2003), "A randomized control trial of household-based flocculant-disinfectant drinking water treatment for diarrhea prevention in rural Guatemala," American Journal of Tropical Medicine and Hygiene 69(4), 411-419.
- Rosen, S. and J.R. Vincent (1999), "Household water resources and rural productivity in Sub-Saharan Africa: A review of the evidence," Harvard Institute for International Development Discussion Paper # 673.
- Schorling, J.B. *et al.* (1990), "Malnutrition is associated with increased diarrhea incidence and duration among children in an urban Brazilian slum." International Journal of Epidemiology 19(3), 728-35.
- Semenza, J. *et al.* (1998), "Water distribution system and diarrheal disease transmission: A case study in Uzbekistan." American Journal of Tropical Medicine and Hygiene 59, 941-6.
- Shordt, K. and S. Cairncross (in press) "It does last! Some findings from the multi-country study of hygiene sustainability," Waterlines.
- Smith, K.R. *et al.* (1999), "How much global ill health is attributable to environmental factors?" Epidemiology 10(5), 573-84.
- Sobsey, M.D. and S. Gundry (2004), World Health Organization International network to promote safe household water treatment and storage draft research agenda. Mimeo revised June, 2004.

- Sobsey, M.D. *et al.* (2003), "Chlorination and safe storage of household drinking water in developing countries to reduce waterborne disease." Water Science and Technology 47(3), 221-8.
- Solow, J. and N. Kirkwood (2002), "Group identity and gender in public goods experiments," Journal of Economic Behavior and Organization 48, 403-412.
- Stanton, B.F. and J.D. Clemens (1987), "An educational intervention for altering water-sanitation behaviors to reduce childhood diarrhea in urban Bangladesh II: A randomized trial to assess the impact of the intervention on hygienic behaviors and rates of diarrhea," American Journal of Epidemiology 125, 292-301.
- Tayeh A. *et al.* (1996), "The impact of health education to promote cloth filters on dracunculiasis prevalence in the northern region, Ghana," Social Science and Medicine 43(8), 1205-11.
- UNEP (United Nations Environment Program) (1998), Sourcebook of alternative technologies for freshwater augmentation in Africa, UNEP: Osaka.
- USAID (United States Agency for International Development) (1996), Environment, health and people, An update on USAID's Environmental Health Project, USAID Environmental Health Project: Arlington.
- USAID (United States Agency for International Development) Micronutrient Program (2004a), Zinc supplementation for the treatment of diarrhea: Moving from research to practice, USAID: Arlington.
- USAID (United States Agency for International Development) Micronutrient Program (2004b), Cost analysis of the national vitamin A supplementation programs in Ghana, Nepal, and Zambia: A synthesis of three studies, USAID: Arlington.
- Varley, R.C.G. *et al.* (1998), "A reassessment of the cost-effectiveness of water and sanitation interventions in programmes for controlling childhood diarrhea," Bulletin of the World Health Organization 76(6), 617-631.
- Vaz, L., and P. Jha (2001), "Note on the health impact of water and sanitation services," World Health Organization Commission on Macroeconomics and Health Working Paper Series Paper #WG5: 23.
- Victora, C.G. *et al.* (1996), "Falling diarrhoea rates in Northeastern Brazil: Did ORT pay a role?" Health Policy and Planning 11(2), 132-41.
- Victora, C.G. *et al.* (2000), "Reducing deaths from diarrhea through oral rehydration therapy," Bulletin of the World Health Organization 78(10), 1246-56.

- Watson, T. (forthcoming), "Public health investments and the infant mortality gap: Evidence from federal sanitation interventions and hospitals on U.S. Indian Reservations," Journal of Public Economics.
- White C.G. *et al.* (2001), "Reduction of illness absenteeism in elementary schools using and alcohol-free instant hand sanitizer," Journal of School Nursing 17(5), 258-65.
- White, G.F. *et al.* (1972), Drawers of water: Domestic water use in East Africa, University of Chicago Press: Chicago.
- WHO (World Health Organization) (2000), Global water supply and sanitation assessment, WHO: Geneva.
- WHO (World Health Organization) (2002), Managing water in the home: Accelerated health gains from improved water supply, WHO: Geneva.
- WHO (World Health Organization) (2002b), World health report 2002, WHO: Geneva.
- WHO (World Health Organization) (2004), "Mozambique mass campaign tests the theory: Vaccines could limit the impact of cholera," press release, available: http://www.who.int/mediacentre/news/releases/2004/pr3/en/print.html. Accessed: 22 October 2004.
- WHO (World Health Organization) collaborative Study Team on the Role of Breastfeeding on the Prevention of Infant mortality (2000), "Effect of breastfeeding on infant and child mortality due to infectious diseases in less developed countries: A pooled analysis," Lancet 355, 451-5.
- Wijk-Sijbesma, C. van (1998), Gender in water resources management, water supply and sanitation: Roles and realities revisited, The Hague: International Water and Sanitation Centre.
- Wilson, J. and G. Chandler (1993), "Sustained improvements in hygiene behaviour amongst village women in Lombok, Indonesia," Transactions of the Royal Society of Tropical Medicine and Hygiene 87, 615-616.
- Wood, S. *et al.* (1998), PHAST step-by-step guide: A participatory approach for the control of diarrheal disease, WHO: Geneva.
- World Bank (2002), Water, sanitation, and hygiene at a glance, World Bank: Washington, DC.
- World Bank (2003), Water A priority for responsible growth and poverty reduction: An agenda for investment and policy change, World Bank: Washington, DC.

World Bank (2004a), "The World Bank Group's Program for Water Supply and Sanitation," Washington, DC: World Bank.

World Bank, (2004b), World Development Report: Making services work for poor people. Washington, DC: World Bank.

Wright, *et al.* (2004), "Household drinking water in developing countries: A systematic review of microbiological contamination between source and point of use," Tropical Medicine and International Health 9(1), 106-17.

Zinc Investigators' Collaborative Group (1999), "Prevention of diarrhea and pneumonia by zinc supplementation in children in developing countries: Pooled analysis of randomized control trials," Journal of Pediatrics 135(6).

Zinc Investigators' Collaborative Group (2000), "Therapeutic effects of oral zinc in acute and persistent diarrhea in children in developing countries: Pooled analysis of randomized control trials," American Journal of Clinical Nutrition 72:1516-22.

Zlotkin, S.H. *et al.* (2005) "Micronutrient sprinkles to control childhood anemia," PLoS Medicine 2(1), 24-28.

Zwarteveen, M. and R. Meizen-Dick (2001), "Gender and property-rights in the commons: Examples of water rights in South Asia," Agriculture and Human Values 18, 11-25.

Tables

Table 1: Transmission routes of water-related diseases

Classification	Transmission route	Examples
Waterborne	Pathogens are ingested in drinking water	 Diarrheal diseases Enteric fevers, such as typhoid Hepatitis A
Water-washed	 Result from having insufficient water for bathing and hygiene purposes Pathogens are incidentally ingested in the course of other activities 	Diarrheal diseasesTrachomaScabies
Water-based	Caused by repeated physical contact with contaminated water, transmission occurs via an aquatic invertebrate host	Guinea wormSchistosomiasis
Water-related insect vector	Transmission occurs via an insect vector which breeds in or near water	Malaria (parasite) and yellow fever (virus)

Sources: White et al. (1972), Feachem et al. (1983), Cairncross and Valdmanis (2004).

Table 2: The cost-effectiveness of interventions to combat diarrheal diseases

Intervention	Reference	Notes	Cost per child diarrhea death averted (1999\$)	Cost per child diarrhea case averted (1999\$)
Standard child health interventions				
Vitamin A supplementation	USAID Micronutrient Program (2004b)	Data from Ghana, Nepal, and Zambia	236 ^a	
Vitamin A supplementation	Loevinsohn et al. (1997)	Data from the Philippines	68-257 a	
Breastfeeding promotion	Martines <i>et al.</i> (1993)	Synthesis of several studies	1,000	71
Improved weaning practices	Martines <i>et al.</i> (1993)	Synthesis of several studies	1,681	
Hypothetical rotavirus vaccine	Martines <i>et al.</i> (1993)	Synthesis of several studies	221-2,215	8
Cholera immunization	Martines <i>et al.</i> (1993)	Synthesis of several studies	3,164	273
ORS	Robberstad et al. (2003)	Data from Tanzania	3,213	
ORS and zinc supplementation	Robberstad et al. (2003)	Data from Tanzania	2,098	
ORS	Varley et al. (1998)	Synthesis of several studies	871	
Environmental health interventions				
Latrine rehabilitation and provision	Meddings et al. (2004) ^b	Data from Afghanistan	2,144	
Hypothetical hardware only	Varley et al. (1998) ^c	Simulation model	41,689	176
Hypothetical hardware and software	Varley et al. (1998)	Simulation model	14,960	64
Hypothetical software only	Varley et al. (1998)	Simulation model	1,599	7
Hygiene education and promotion	Martines <i>et al.</i> (1993)	Synthesis of several studies		16

Notes:

⁽a) Randomized control trial estimates that include deaths averted from measles.

⁽b) Case-control study using data collected in hospitals.(c) Varley *et al.* (1998) model provision of tubewells and taps to a peri-urban community. Costs in rural areas would be higher.